Author's response to reviews

Title: Assessing record linkage between health care and Vital Statistics databases using deterministic methods

Authors:

Bing BL Li (LIB@UCALGARY.CA)
Hude HQ Quan (hquan@ucalgary.ca)
Andrew AF Fong (andrew.fong@calgaryhealthregion.ca)
Mingshan ML Lu (lu@ucalgary.ca)

Version: 5 Date: 31 March 2006

Author's response to reviews: see over
Dear Editors:

Following your instructions, we have reformatted our manuscript. On page 1, we checked the authors list and made authors’ name identical with those in the submission system: “Bing BL Li\textsuperscript{1}, Hude HQ Quan\textsuperscript{1,2,3,*}, Andrew AF Fong\textsuperscript{1}, Mingshan ML Lu\textsuperscript{1,3,4}”. On page 19, we placed two figure legends together. We appreciate your work on our paper and look forward to hearing from you.

Yours sincerely,

Bing Li
(on behalf of all co-authors)